Author's response to reviews

Title: All-arthroscopic versus mini-open repair of small or moderate-sized rotator cuff tears: A protocol for a randomized trial [NCT00128076] - A study protocol

Authors:

Joy C MacDermid (macderj@mcmaster.ca)
Richard Holtby (richard.holtby@sw.ca)
Helen Razmjou (helen.razmjou@sw.ca)
Dianne Bryant (jackowdm@mcmaster.ca)
Joints Canada (jointscanada@jointscanada.ca)

Version: 4 Date: 21 February 2006

Author's response to reviews: see over
Title & Abstract - Please ensure these are identical in the manuscript and in the submission system.

Checked.

Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.

Checked.

Title - Please use sentence case in the title, both in the manuscript file and the submission system (i.e. remove all unnecessary capitals).

Unnecessary capitals in the title were removed.

Affiliations - Each affiliation should be written in full and be provided in the following format: department, institute, city, country. This may mean repeating institution addresses if authors are from different departments within an institution. Please do not use full capitals for country names, and please do not include full-stops in abbreviated country names.

Country added.

Abstract - Abstracts must be structured into Background, Methods/Design, Discussion. Please remember to also update the Abstract details on the submission page.

Abstract on the submission page has been updated.

Underlining - Please do not underline any section headings

All underlining has been removed. Subheadings italicized instead.